From no-reply@ftp.scri.fsu.edu Tue Dec  7 22:19:56 1993
Received: from ftp.scri.fsu.edu by spica.npac.syr.edu (4.1/I-1.98K)
	id AA25342; Tue, 7 Dec 93 22:19:44 EST
Received: by ftp.scri.fsu.edu id AA22718
  (5.65c/IDA-1.4.4 for paulc@npac.syr.edu); Tue, 7 Dec 1993 22:19:37 -0500
Date: Tue, 7 Dec 1993 22:19:37 -0500
Message-Id: <199312080319.AA22718@ftp.scri.fsu.edu>
Precedence: bulk
Note: bulletin board software written by PG at LANL (8/91, 12/92)
From: no-reply@ftp.scri.fsu.edu  (send mail ONLY to hep-lat)
Reply-To: hep-lat@ftp.scri.fsu.edu
To: paulc@npac.syr.edu
Subject: RE: get 9312025
Status: RO

%Paper: hep-lat/9312025
%From: Jacek Wosiek <wosiek@ztc386a.if.uj.edu.pl>
%Date: Sun, 5 Dec 93 12:17:01 cet


% This paper contains four figures. First three are in LATEX and form
% a single LATEX document together with the article. Figure 4 is the
% Postcsript output from the GRAPHER package, an is appended after
% the \end{document} command. Copy it into a separate file and send to
% your Postscript printer.
%
\documentstyle [12pt]{article}

\begin{document}
\begin{thebibliography}{99}
\bibitem{huang} K. Huang, {\em Statistical Mechanics}, John
Wiley \& Sons, Inc., 1963.
\bibitem{ons} L.Onsager, Phys. Rev. {\bf 65}, 117 (1944).
\bibitem{kauf} B. Kaufmann, Phys. Rev. {\bf 76}, 1232 (1949).
%B. Kaufman and L. Onsager, {\em ibid}, 1244.
\bibitem{wu} F. Y. Wu, Pev. Mod. Phys. {\bf 54}, 235 (1982).
\bibitem{bax} R. J. Baxter, {\em Exactly Solved Models in Statistical
Mechanics}, Academic Press, 1982.
\bibitem{has} C. F. Baillie, R. Gupta, K. A. Hawick and G. S. Pawley,
Phys. Rev. {\bf B45}, 10438 (1992).
\bibitem{bha} G. Bhanot, M. Creutz and S. Lacki,
 Phys. Rev. Lett. {\bf 69}, 1841 (1992).
\bibitem{stod} L. Stodolsky and J. Wosiek, MPI-Ph/92-97, Nucl. Phys. {\bf B}
{\em in press}; {\em see also} J. Wosiek, Nucl. Phys. {\bf B} (Proc. Suppl.)
{\bf 30}, 239 (1993).
\bibitem{KT} J. M. Kosterlitz and D. J. Thouless, J. Phys. {\bf C6},
1181 (1973);
J. M. Kosterlitz, {\em ibid}, 1046 (1974).
\bibitem{pin} M. Hasenbusch, M. Marcu and K. Pinn, Nucl. Phys. {\bf B}
(Proc. Suppl.) {\bf 26}, 598 (1992).
\bibitem{gup} R. Gupta, C. F. Baillie, Phys. Rev. {\bf B45}, 2883 (1992-II).
\end{thebibliography}


\vspace{10mm}
\centerline{\Large\bf Locating analytically critical temperature}
\centerline{\Large\bf in some statistical systems }
\vspace*{1cm}
\begin{center}
{\large J. Wosiek}
\\
\vspace*{1.5mm}
Institute of Physics, Jagellonian University \\
Reymonta 4, 30-059 Cracow, Poland
\end{center}
\vspace*{1.5cm}
\hspace*{\fill} \fbox{To my Father} \newline
\vspace*{1.5cm}
\begin{abstract}
We have found a simple criterion which allows for the straightforward
determination of the order-disorder critical temperatures.
The method reproduces exactly  results known for the two dimensional Ising,
Potts and $Z(N<5)$ models. It also works for the Ising model on the
triangular lattice. For systems which are not selfdual our proposition remains
an unproven conjecture.
It predicts $\beta_c=0.2656...$ for the two coupled
layers of Ising spins. Critical temperature of the three dimensional
Ising model is related to the free energy of the two layer Ising system.
\end{abstract}
\vspace*{2cm}
PACS number: 64.60.Cn \newline
hep-lat/9312025
\newline
%\vspace*{2cm}
TPJU-23/93 \newline
November 1993 \newline
\newpage

 Classical method of determining critical temperature in statistical physics
 consists
of locating singularities of the largest eigenvalue of the corresponding
transfer matrix
${\cal T}$. To this end one has to study the high power of ${\cal T}$, e.g.
${\cal T}^L$ with
$L$ being the linear size of the system. This amounts to investigate the $d$
dimensional
euclidean system in its full complexity. On the other hand it is conceivable
that the
information about the phase transition is also encoded in all other
eigenvalues, i.e. that
the whole spectrum of the transfer matrix is sensitive to its location.
This observation is confirmed
by the famous example of the two dimensional Ising model where {\em all}
eigenvalues show the extremal behaviour at $\beta=\beta_c$ \cite{huang}.
Therefore we propose to search for the extremum of some simpler
function which characterizes the system, which however is not
dominated solely by the largest eigenvalue. The advantage of such an
approach is that one may infer a nontrivial information from
"characteristic" functions which are much easier to calculate.
To be specific we propose to study the following characteristic function
of a $d$ dimensional system
\begin{equation}
\rho(\beta) = \lim_{L\rightarrow\infty} \left( {(Tr {\cal T})^2 \over
Tr {\cal T}^2 }
\right )^{{1\over L^{d-1}}} ,  \label{rho}
\end{equation}
and analogous higher, moments of the transfer matrix.
Following simple properties of $\rho$ can be easily proven.
\begin{eqnarray}
a)  & \rho(0)=1, \nonumber \\
b)  & \rho(\infty)=1, \\
c)  &  \rho(\beta) \ge 1 .     \nonumber
\label{abc}
\end{eqnarray}
To see $a)$ consider the normalized second moment at finite $L$
\begin{equation}
r_L(\beta)={(Tr {\cal T})^2 \over Tr {\cal T}^2 }= {(Z_1)^2 \over Z_2},
\label{rl}
\end{equation}
where $Z_1$ and $Z_2$ are the partition functions of the $d-1$ dimensional
system and of the two coupled $d-1$ dimensional
systems respectively. For simplicity we shall use the
terminology of the d=2 spin system. Hence $Z_1$
and $Z_2$ describe one dimensional
spin chain and two coupled chains of spins. Periodic boundary conditions are
implied
in both directions, even for the single chain. With these definitions $a)$
follows immediately from the observation that
$Z_2(\beta=0)={\cal N}_2=({\cal N}_1)^2$, where
${\cal N}_1, {\cal N}_2$ denote the total number of microscopic states of
single and double
chain.  At low temperatures $(\beta \rightarrow \infty)$ fully ordered states
dominate, hence
\begin{equation}
\rho(\beta=\infty)=\lim_{L\rightarrow \infty} g^{1\over L^{d-1}} =1.
\end{equation}
Where $g$ denotes degeneracy factor of the ordered states.
The last equality requires finite $g$ (and $d>1$),
 therefore $b)$ holds only for
systems with the discrete internal symmetry. Finally the
property $c)$ follows directly from the positivity of the
eigenvalues of the transfer matrix since
\begin{equation}
r_L(\beta)= {
              \sum_{\alpha,\beta} \lambda_{\alpha} \lambda_{\beta}
           \over \sum_{\alpha} \lambda_{\alpha}^2
        } > 1  .
\label{pos}
\end{equation}

Characteristic function, Eq.(\ref{rho}), is sensitive to the whole
spectrum of the transfer matrix. However in view of our earlier
discussion supplemented with the properties
$a) - c)$ it is natural to expect that the maximum of $\rho$ occurs
at the phase transition
point,
\begin{equation}
\beta_{max}=\beta_c   .   \label{prop}
\end{equation}
Surprisingly this simple proposition is true in many,
sometimes nontrivial, cases.
We shall discuss them in the order of increasing complexity.

{\em Two dimensional Ising model.}
For $d=2$,  calculation of $\rho(\beta)$ is easily reduced to
solving a straightforward two spin problem.
Indeed the partition functions $Z_i (i=1,2)$ are readily written as
\begin{equation}
Z_i=Tr (T_i)^L ,   \label{smallt}
\end{equation}
where $T_i$ are the transfer matrices propagating one/two spins
{\em horizontally}, c.f. Fig.1.

Explicitly for Ising spins
\begin{eqnarray}
T_1=\left( \begin{array}{cc}
             x^2 & x   \\
             x   & x^2 \\
           \end{array} \right),
T_2=\left(\begin{array}{cccc}
        x^4 & x^2 & x^2 & x^2 \\
        x^2 & x^2 & 1   & x^2 \\
        x^2 & 1   & x^2 & x^2 \\
        x^2 & x^2 & x^2 & x^4 \\
         \end{array}       \right), x=e^\beta.
\end{eqnarray}
We have chosen the interaction energy
\begin{equation}
E(s)=-\sum_{<nm>} \delta_{s_n, s_m}, \label{ip}
\end{equation}
and $s_n=\pm 1$.
In the thermodynamical limit the largest eigenvalues of $T_i$ dominate and
we get
\begin{equation}
\rho_{Ising2}(\beta)=  { t_{1max}^2 \over t_{2max} } = { x^2(x+1)^2
                              \over
\frac{1}{2}(x^2+1)^2+\sqrt{\frac{1}{4}(x^2-1)^4+x^2(x^2+1)^2}} .  \label{i2}
\end{equation}
Since the $L\rightarrow \infty$ limit was already performed, $\rho(\beta)$,
 as given
by Eq.(\ref{i2}), is the nontrivial characteristic of the infinite system.
As conjectured
$\rho(\beta)$ has a single maximum at $x_c=1+\sqrt{2}$ which corresponds
to the famous
Onsager value
\cite{ons,kauf} \footnote{Note the difference by a factor of 2 which
is caused by
our choice of the Potts-like interactions in Eq.(\ref{ip}).} .

{\em Potts and equivalent $Z(N)$ models.}  The energy of the
$q$ state Potts model is given by Eq.(\ref{ip})
with $s_n$ assuming $q$ different values. Transfer matrices for one
and two spin system are again simple
\begin{eqnarray}
<s|T_1|s'>&=&\exp{\beta (\delta_{s,s'}+1)},\\  \nonumber
<s_1,s_2|T_2|s_1',s_2'>&=&
\exp{\beta(\delta_{s_1,s_2}+\delta_{s_1,s_1'}+\delta_{s_2,s_2'}
+\delta_{s_1',s_2'})} . \label{Tpotts}
\end{eqnarray}
For $q=3$ the diagonalization is tractable \footnote{All algebraic
calculations were done
with the aid of MATHEMATICA.} and the final result for the characteristic
function
reads
\begin{eqnarray*}
\lefteqn{\rho_{Potts3}(\beta)=} \\
 & & {2x^2(x+2)^2 \over
x^4+3x^2+2x+3+\sqrt{x^8+2x^6-4x^5+27x^4+28x^3+6x^2+12x+9}  }
\end{eqnarray*}
Again it has a single maximum at $x_c=1+\sqrt{3}$ which agrees with the
known location
of the transition temperature \cite{wu}.

For $q=4,5$
we have used standard numerical methods
to diagonalize transfer matrices. In both cases
$\rho(x)$ has a sigle maximum located at $x_c=1+\sqrt{q}$ in accord with
known results.
For $q > 4$ the transition is  first order \cite{wu} and consequently
our method seems
to apply to both kinds of transitions. We have not attempted algebraic
diagonalization
of $T_i$ for arbitrary $q$.
Low $N$ $(N<5)$ $Z(N)$ (clock) models are equivalent to Potts3 (N=3) and
Ising (N=4)
 systems. Not surprisingly the maximum of $\rho(\beta)$ again agrees
with known results.
We obtain $x^{Z(3)}_c=(1+\sqrt{3})^{2\over 3}$ and $x^{Z(4)}_c=1+\sqrt{2}$
within the accuracy of our numerical procedures.

All previously considered systems were selfdual.
Therefore one may justifiably wonder if our principle is not yet
another manifestation
of the selfduality. Next example demonatrates that the ``maximum rule'',
Eq.(\ref{prop}), is at least more general than the simple duality.

{\em Ising model on a triangular lattice.}
Transfer matrices $T_{1,2}$ \footnote{From now on we use the standard
Ising energy $E(s)=-\sum_{<ij>} s_i s_j$.}
(see Fig.2.)
\begin{eqnarray}
<s_1|T_1|s_1'>&=&\exp{\beta(2 s_1 s_1'+1)},\\  \nonumber
<s_1,s_2|T_2|s_1',s_2'>&=&
\exp{\beta(s_1 s_1'+s_2 s_2'+s_1 s_2 +s_1's_2'+s_1 s_2'+s_2 s_1')}, \nonumber
\end{eqnarray}
can be simply diagonalized. We get
\begin{eqnarray}
t_1^{max}&=&x^3+x^{-1}, \\
t_2^{max}&=&{1\over 2 x^2}(3+x^8+\sqrt{x^{16}-2x^8+16 x^4+1}).
\end{eqnarray}
The maximum of the characteristic function is located at $x_{max}=
3^{1\over4}$.
This agrees with the critical temperature
first derived for this system by Onsager \cite{ons}.
One should remember however that Ising models on triangular and hexagonal
lattices
are interrelated via the duality and star-trangle relations,
hence effectively there exists a symmetry which determines transition
points in both systems \cite{bax}. Our maximum rule could in principle
be a consequence of such a symmetry in this case.
Next application provides more stringent test of this
possibility.

{\em Two layer Ising model.} This system consists of the two planes of
Ising spins coupled
by the nearest neighbour ferromagnetic interaction, also along the vertical
(between the
planes) direction. Periodic boundary conditions  are assumed in all three
directions, which
amounts to doubling the strengh of interaction between the planes.
The system in not selfdual and to our knowledge no other more complicated
symmetries are known.
 Consequently its critical temperature was never derived.  On the contrary
our method provides relatively simple analytic predictions for $\beta_{max}$.
We proceed analogously to the previous cases. Transfer matrices $T_{1,2}$
propagate now states of two and four spins respectively (see Fig.3).
$T_1$ reads
\begin{equation}
<s_1s_2|T_1|s_1's_2'>=\exp{\beta(s_1s_2+s_1s_1'+s_2s_2'+s_1's_2'+2)},
\;\;\;s_i=\pm 1,
\end{equation}
and its largest eigenvalue is
\begin{equation}
t_1^{max}={x^2\over 2}(x^4+2+x^{-4}+\sqrt{x^8+14+x^{-8}}). \label{t1}
\end{equation}
Matrix elements of $T_2$ are
\begin{eqnarray*}
<s_1s_2s_3s_4|T_2|s_1's_2's_3's_4'>=\exp{\beta(s_1s_2+s_2s_3+
s_3s_4+s_4s_1)}\\
\exp{\beta(s_1s_1'+s_2s_2'+s_3s_3'+s_4s_4')}
\exp{\beta(s_1's_2'+s_2's_3'+s_3's_4'+s_4's_1')}.
\end{eqnarray*}
Diagonalisation of this $16\times16$ matrix is simplified noting that
$T_2$ conserves
the $U$-parity, $[T_2,U]=0, U=\prod_{i=1}^4 \sigma_i^x$.
The largest eigenvalue
belongs to the $U=+1$ sector. Final expression is little more complicated
\begin{eqnarray}
t_2^{max}&=&{(1+x^4)^2 \over 4 x^{12}} q_3  +  {1+x^8 \over
4x^{12}}\sqrt{q_1}
+{1+x^4 \over 2\sqrt{2} x^{12}} \sqrt{q_2+(1+x^8) q_3 \sqrt{q_1} }, \\
\label{dlay}
q_1(x)&=&x^{32}-4x^{24}+70x^{16}-4x^8+1,\\ \nonumber
q_2(x)&=&x^{40}-2x^{36}+5x^{32}+26x^{24}+4x^{20}+26x^{16}+5x^8-2x^4+1,\\
\nonumber
q_3(x)&=&x^{16}-2x^{12}+6x^8-2x^4+1. \nonumber
\end{eqnarray}
Resulting characteristic function $\rho(\beta)$ is shown in Fig.4.
It has the single maximum
located at
\begin{equation}
\beta_{max}=0.2656...\;\; .   \label{betac}
\end{equation}
 According to our proposition, Eq.(\ref{prop}), this gives the
transition temperature of the two layer
Ising system.
For comparison:
$\beta_c^{Ising2}\simeq 0.4407...$ and
$\beta_c^{Ising3}\simeq 0.221652(3)$ \cite{has}.
Confronting this number
with the results from MC simulations
would provide the crucial test of our hypothesis. It would be also very
interesting to search
for the "generalized selfduality" - the invariance which would
assure existence of a single maximum at the transition point.
An attractive possibility is to use the characteristic function
given by Eqs.(\ref{t1},\ref{dlay})
 to {\em define } such a mapping. This assumption has
many verifiable consequences. For example, all higher normalized moments of
${\cal T}$ should respect the same symmetry.

{\em Three dimensional Ising model.}  According to our proposition the
transition temperature of
the $d$ dimensional system is determined by the $\beta$ dependence
of the free energies $\beta F_{1,2}=-\log{Z_{1,2}}$ of the corresponding
$d-1$
dimensional
systems. In particular, the problem of finding $\beta_c$ for
the three dimensional Ising
model would be reduced to finding the free energy of the two coupled
layers of
the Ising spins. Indeed in this case the full transfer matrix ${\cal T}$
propagates the whole plane of spins, say, vertically, while the reduced
transfer
 matrices $T_{1,2}$ propagate one (two) rows
horizontally. No analytic solution of this system exists up to date
\footnote{In our previous application we have
derived the transition temperature only.}. However exact expressions
for the complete partition functions $Z_{1,2}(\beta,L)$
{\em in the finite volume} are available for not so large $L$
\cite{bha,stod}.
We have therefore calculated the exact locations
$\beta_{max}(L)$ of the ratio $r_L(\beta)$, c.f. Eq. (\ref{rl}),
which define the pseudocritical temperature
at finite volume, c.f. Table I. \newline
\begin{table}
\begin{tabular}{c|ccccc} \hline\hline
 $L$         &     3  &     4  &     5  &     6  &  7     \\  \hline\hline
 $\beta_{max}(L)$& 0.3317 & 0.3067 & 0.2938 & 0.2859 & 0.2698 \\ \hline\hline
\end{tabular} \newline
\caption{Table I. Volume dependence of the pseudocritical temperature
for the three dimensional Ising model.}
\end{table}
In the thermodynamical limit $\beta_{max}(L)$ should converge to the
true transition
temperature. Even though the available range of $L$ values is rather
limited one sees
the proper trend in the $L$ dependence.  Our values definitely move toward
$\beta_c^{Ising3}$ which was quoted above. One needs larger sizes to
test quantitatively the $L$ dependence against the finite size
scaling predictions.

Variety of approximate methods (Monte Carlo, high temperature expansion)
can  be  also employed to test predictions of Eq.(\ref{prop}),
 in this case.

{\em Limits of applicability.} We have also investigated situations where
the maximum rule does not work.
The regularity emerging from this study indicates that the method does
not apply to
 systems with more than two phases.
We have calculated the characteristic function for the variety of models
with the intermediate Kosterlitz-Thouless (KT) \cite{KT} phase.
The maximum always occured inside the KT region. This phenomenon was found
for $Z(N)$ models with N=5-19, and for the icosahedron and dodecahedron
models
in two dimensions. For the O(2) model $\beta_{max}=1.35...$ well above the
known MC estimate
for the transition between the disordered and KT phases $\beta_c=1.1197(5)$
\cite{pin,gup}.
Interpretation of $\beta_{max}$ located inside the KT phase remains an
open and interesting problem.  On the other hand in the two dimensional
O(3) model $\rho(\beta)$
does not have any maximum in accord with the common wisdom about the lack of
a phase transition in this model. Note that for O(2) and O(3) models the
property c) is not
satisfied and indeed calculated $\rho(\infty) > 1$.
Nevertheless the characteristic function
{\em has} a maximum for $O(2)$ while it is monotonic for $O(3)$.
This parallels the difference
of the phase structures of both models.

{\em Summary.} We have found a surprisingly simple criterion for locating
order-disorder
transition. It is exact for selfdual models.
The method allows for the analytic calculation
in variety of more complicated systems. In particular we give the analytic
estimate of the critical
temperature of the two layer Ising system. Monte Carlo check of this
prediction should be the first step towards more
advanced applications.

Our proposition reduces determination of the critical temperature
of the three dimensional Ising model  to finding
the $\beta$ dependence of the free energy of the two coupled planes
of Ising spins.

 While we are lacking the complete proof of our
hypothesis in general case, many approximated methods can be employed
to test it in
specific applications. High temperature expansion is one
interesting possibility.

One can reformulate the maximum principle in other equivalent ways.
Differentiating
the logarithm of Eq.(\ref{rl}) gives as the condition for the maximum
\begin{equation}
u_2(\beta_c)=u_1(\beta_c),   \label{us}
\end{equation}
where $u_{1,2}$ denotes the density of the internal energy for the
one (two) layer
system.  Analogous relations follow from applying our proposition to
higher moments
of transfer matrix. They all say that at the bulk ($d=3$ say) critical
temperature internal energies
of the interacting and noninteracting planes are equal. According to
our earlier
discussion this statement follows from duality for the two dimensional
Ising and Potts models with planes replaced by the spin chains.
Perhaps the most interesting formulation of Eq.(\ref{us}) for higher moments
of the transfer matrix results in the limit of infinite number of planes.
Then we recover the original two dimensional system and our criterion reads
\begin{equation}
u_{d}(\beta_c^{(d)})=u_{d-1}(\beta_c^{(d)}) \label{uinf},
\end{equation}
where $u_{d}$ denotes the internal energy density of the $d$
dimensional system, and $\beta_c^{(d)}$ corresponds
to its crtitical temperature. As emphasized before, this statement follows
from selfduality in the case of the two dimensional Ising model.
It can be also
checked directly. Indeed
\begin{equation}
u_{2}(\beta_c)=u^{Onsager}_{|2\beta=\log{(1+\sqrt{2})}}=-\sqrt{2}=
u_{1}(\beta_c),
\end{equation}
where
$u^{Onsager}(\beta)=-ctgh(2\beta)\left[1+
{2\kappa^{'}\over \pi} K_1(\kappa)\right]$, $\kappa'=2\tanh{(2\beta)}^2-1$,
$\kappa^2+\kappa'^2=1$, $K_1$ is the elliptic function of the
first kind \cite{huang}, and $u_{1}(\beta)=-\tanh{(\beta)}-1$.
For $d>2$ Eq.(\ref{uinf}) remains unproven similarly
to Eq.(\ref{prop}).


To conclude, there are many unanswered questions and much more work is to be
done, but we feel that it is certainly worth
to undertake this effort.
\vspace*{1cm}

This work was triggered by L. Stodolsky interest in the feasibility
of calculating low moments of the transfer matrix. I would like to thank him
for numerous and stimulating discussions. I also thank E. Seiler and
A. Sokal for instructive discussions.



\section*{Figure Captions}
{\bf Fig.1.} Construction of the transfer matrices ${\cal T}$ and $T_2$
for the two dimensional Ising model. Periodic boundary conditions
are understood. \newline
{\bf Fig.2.} Same as Fig.2 but for the triangular lattice. Bonds
impiled by the periodic boundary conditions are shown explicitly.
\newline
{\bf Fig.3.} Same as Fig.3 but for the two layer Ising model.
\newline
{\bf Fig.4.} Characteristic function for the two layer Ising model.
\newpage
\begin{figure}
\begin{picture}(100,270)(-50,-30)
\multiput(20,50)(36,0){4}{\circle*{5}}
\multiput(20,86)(36,0){4}{\circle*{5}}
\multiput(145,68)(6,0){3}{\circle*{2}}
\multiput(170,50)(0,36){2}{\circle*{5}}
\put (20,50) {\line(124,0){124} }
\put (20,86) {\line(124,0){124} }
\put (50,120) { $ T_2$  }
\put(80,125){\vector(1,0){40}}
\multiput(20,50)(36,0){4}{\line(0,36){36} }
\put(170,50) {\line(0,36){36} }
\put(210,45) {${\cal T}$}
\put(214,60){\vector(0,1){20}}
\put(128,-100){{\Large Fig.1}}
\end{picture}
%\caption{Fig.1.}
\end{figure}
\pagebreak
\newpage
\begin{figure}
\begin{picture}(100,270)(-50,-30)
\multiput(20,50)(36,0){4}{\circle*{5}}
\multiput(20,86)(36,0){4}{\circle*{5}}
\multiput(20,122)(36,0){4}{\circle{5}}
\multiput(145,68)(6,0){3}{\circle*{2}}
\multiput(170,50)(0,36){2}{\circle*{5}}
\multiput(206,50)(0,36){3}{\circle{5}}
\put(170,122){\circle{5}}
\multiput(170,50)(0,36){2}{\line(1,0){36}}
\multiput(170,50)(0,36){2}{\line(1,1){36}}
\put (20,50) {\line(124,0){124} }
\put (20,86) {\line(124,0){124} }
\put (50,156) { $ T_2$  }
\put(80,161){\vector(1,0){40}}
\multiput(20,50)(36,0){4}{\line(0,36){72} }
\multiput(20,86)(36,0){3}{\line(1,1){36} }
\multiput(20,50)(36,0){3}{\line(1,1){36} }
\put(128,50) {\line(1,1){16} }
\put(128,86) {\line(1,1){16} }
\put(170,50){\line(0,36){72} }
\put(246,45) {${\cal T}$}
\put(250,60){\vector(0,1){20}}
\put(128,-100){{\Large Fig.2}}
\end{picture}
%\caption{Fig.1.}
\end{figure}
\pagebreak
\newpage
\begin{figure}
\begin{picture}(100,270)(-50,-30)
\multiput(20,50)(36,0){4}{\circle*{5}}
\multiput(20,86)(36,0){4}{\circle*{5}}
\multiput(32,68)(36,0){4}{\circle*{5}}
\multiput(32,104)(36,0){4}{\circle*{5}}
\multiput(150,76)(6,0) {3}{\circle*{2}}
\multiput(170,50)(0,36){2}{\circle*{5}}
\multiput(182,68)(0,36){2}{\circle*{5}}
\put (20,50) {\line(124,0){124} }
\put (20,86) {\line(124,0){124} }
\multiput(32,68)(0,36){2}{\line(1,0){124}}
\put (50,138) { $ T_2$  }
\put(80,143){\vector(1,0){40}}
\multiput(20,50)(36,0){4}{\line(0,1){36}}
\multiput(20,50)(36,0){4}{\line(2,3){12}}
\multiput(20,86)(36,0){4}{\line(2,3){12}}
\multiput(32,68)(36,0){4}{\line(0,1){36}}
\multiput(170,50)(12,18){2}{\line(0,1){36}}
\multiput(170,50)(0,36){2}{\line(2,3){12}}
\put(210,45) {${\cal T}$}
\put(214,60){\vector(0,1){20}}
\put(128,-100){{\bf {\Large Fig.3}}}
\end{picture}
%\caption{Fig.3.}
\end{figure}
\end{document}
%%%%%%%%%%%%%%%%%%%Fig.4.%%%%%%%%%%%%%%%%%%%%%%%%%%
% PostScript Driver
% Copyright (C) 1988 Golden Software, Inc.
gsave
newpath
.072 .072 scale
/gsdict 4 dict def
gsdict begin
/M /moveto load def
/D /lineto load def
/S /stroke load def
/W /setlinewidth load def
S
3 W
S
2050 1700 M 6250 1700 D S
2050 1700 M 2050 1779 D S
1892 1470 M 1973 1545 D S
1973 1535 M 1947 1560 D 1918 1560 D 1892 1535 D 1892 1480 D 1918 1455 D 1947
1455 D 1973 1480 D 1973 1535 D S
2023 1455 M 2037 1455 D 2037 1470 D 2023 1470 D 2023 1455 D S
2102 1470 M 2182 1545 D S
2182 1535 M 2157 1560 D 2128 1560 D 2102 1535 D 2102 1480 D 2128 1455 D 2157
1455 D 2182 1480 D 2182 1535 D S
3450 1700 M 3450 1779 D S
3293 1470 M 3372 1545 D S
3372 1535 M 3347 1560 D 3318 1560 D 3293 1535 D 3293 1480 D 3318 1455 D 3347
1455 D 3372 1480 D 3372 1535 D S
3422 1455 M 3438 1455 D 3438 1470 D 3422 1470 D 3422 1455 D S
3502 1470 M 3517 1455 D 3557 1455 D 3582 1480 D 3582 1495 D 3568 1510 D 3517
1510 D 3502 1495 D 3502 1560 D 3582 1560 D S
4850 1700 M 4850 1779 D S
4708 1535 M 4733 1560 D 4733 1455 D S
4823 1455 M 4838 1455 D 4838 1470 D 4823 1470 D 4823 1455 D S
4903 1470 M 4983 1545 D S
4983 1535 M 4958 1560 D 4928 1560 D 4903 1535 D 4903 1480 D 4928 1455 D 4958
1455 D 4983 1480 D 4983 1535 D S
6250 1700 M 6250 1779 D S
6108 1535 M 6132 1560 D 6132 1455 D S
6223 1455 M 6238 1455 D 6238 1470 D 6223 1470 D 6223 1455 D S
6303 1470 M 6318 1455 D 6358 1455 D 6383 1480 D 6383 1495 D 6368 1510 D 6318
1510 D 6303 1495 D 6303 1560 D 6383 1560 D S
4101 1152 M 4109 1218 D 4117 1268 D 4126 1302 D 4134 1327 D 4151 1360 D 4167
1377 D 4184 1385 D 4159 1377 D 4142 1360 D 4126 1327 D 4117 1302 D 4109 1268 D
4101 1218 D 4092 1152 D S
4201 1385 M 4217 1368 D 4217 1343 D 4209 1327 D 4201 1318 D 4184 1310 D S
4117 1252 M 4126 1227 D 4134 1218 D 4151 1210 D 4167 1210 D 4192 1218 D 4201
1227 D 4209 1243 D 4209 1268 D 4201 1285 D 4184 1302 D 4151 1310 D 4184 1310 D
4209 1318 D 4217 1327 D 4226 1343 D 4226 1368 D 4217 1377 D 4201 1385 D 4184
1385 D S
4151 1310 M 4176 1302 D 4192 1285 D 4201 1268 D 4201 1243 D 4192 1227 D 4184
1218 D 4167 1210 D S
2050 1700 M 2050 5900 D S
2050 1700 M 2128 1700 D S
1595 1662 M 1675 1738 D S
1675 1728 M 1650 1753 D 1620 1753 D 1595 1727 D 1595 1672 D 1620 1648 D 1650
1648 D 1675 1673 D 1675 1728 D S
1725 1648 M 1740 1648 D 1740 1663 D 1725 1663 D 1725 1648 D S
1830 1648 M 1885 1703 D 1885 1738 D 1870 1753 D 1820 1753 D 1805 1738 D 1805
1713 D 1820 1703 D 1870 1703 D 1885 1713 D S
2050 2992 M 2128 2992 D S
1610 3020 M 1635 3045 D 1635 2940 D S
1725 2940 M 1740 2940 D 1740 2955 D 1725 2955 D 1725 2940 D S
1805 3030 M 1820 3045 D 1870 3045 D 1885 3030 D 1885 3020 D 1860 2995 D 1885
2965 D 1885 2955 D 1870 2940 D 1820 2940 D 1805 2955 D S
2050 4284 M 2128 4284 D S
1610 4312 M 1635 4337 D 1635 4232 D S
1725 4232 M 1740 4232 D 1740 4247 D 1725 4247 D 1725 4232 D S
1805 4337 M 1885 4337 D 1820 4232 D S
2050 5577 M 2128 5577 D S
1595 5604 M 1595 5614 D 1610 5629 D 1660 5629 D 1675 5614 D 1675 5589 D 1595
5539 D 1595 5524 D 1675 5524 D S
1725 5524 M 1740 5524 D 1740 5539 D 1725 5539 D 1725 5524 D S
1820 5604 M 1845 5629 D 1845 5524 D S
1047 4892 M 1056 4867 D 1064 4858 D 1081 4850 D 1097 4850 D 1122 4858 D 1139
4883 D 1147 4908 D 1147 4933 D 1139 4950 D 1131 4958 D 1114 4967 D 1097 4967 D
1072 4958 D 1056 4933 D 1047 4908 D 1014 4792 D S
1097 4850 M 1114 4858 D 1131 4883 D 1139 4908 D 1139 4942 D 1131 4958 D S
1097 4967 M 1081 4958 D 1064 4933 D 1056 4908 D 1022 4792 D S
1264 4792 M 1247 4808 D 1231 4833 D 1214 4867 D 1206 4908 D 1206 4942 D 1214
4983 D 1231 5017 D 1247 5042 D 1264 5058 D 1256 5058 D 1239 5042 D 1222 5017 D
1206 4983 D 1197 4942 D 1197 4908 D 1206 4867 D 1222 4833 D 1239 4808 D 1256
4792 D 1264 4792 D S
1314 4792 M 1322 4858 D 1331 4908 D 1339 4942 D 1347 4967 D 1364 5000 D 1381
5017 D 1397 5025 D 1372 5017 D 1356 5000 D 1339 4967 D 1331 4942 D 1322 4908 D
1314 4858 D 1306 4792 D S
1414 5025 M 1431 5008 D 1431 4983 D 1422 4967 D 1414 4958 D 1397 4950 D S
1331 4892 M 1339 4867 D 1347 4858 D 1364 4850 D 1381 4850 D 1406 4858 D 1414
4867 D 1422 4883 D 1422 4908 D 1414 4925 D 1397 4942 D 1364 4950 D 1397 4950 D
1422 4958 D 1431 4967 D 1439 4983 D 1439 5008 D 1431 5017 D 1414 5025 D 1397
5025 D S
1364 4950 M 1389 4942 D 1406 4925 D 1414 4908 D 1414 4883 D 1406 4867 D 1397
4858 D 1381 4850 D S
1497 4792 M 1514 4808 D 1531 4833 D 1547 4867 D 1556 4908 D 1556 4942 D 1547
4983 D 1531 5017 D 1514 5042 D 1497 5058 D 1489 5058 D 1506 5042 D 1522 5017 D
1539 4983 D 1547 4942 D 1547 4908 D 1539 4867 D 1522 4833 D 1506 4808 D 1489
4792 D 1497 4792 D S
2078 2154 M 2086 2193 D 2095 2233 D 2103 2272 D 2111 2312 D 2119 2351 D 2128
2391 D 2135 2430 D 2144 2470 D 2152 2509 D 2160 2549 D 2168 2588 D 2176 2628 D
2184 2667 D 2191 2707 D 2200 2747 D 2208 2786 D 2215 2826 D 2223 2866 D 2231
2905 D 2239 2945 D 2247 2984 D 2254 3024 D 2262 3064 D 2270 3104 D 2278 3143 D
2285 3183 D 2293 3222 D 2301 3262 D 2308 3302 D 2316 3342 D 2324 3381 D 2331
3421 D 2339 3461 D 2347 3500 D 2355 3540 D 2363 3579 D 2371 3619 D 2378 3659 D
2386 3698 D 2394 3738 D 2402 3777 D 2410 3817 D 2418 3856 D 2426 3896 D 2434
3935 D 2442 3975 D 2450 4014 D 2459 4054 D 2467 4093 D 2476 4133 D 2484 4172 D
2493 4212 D 2502 4251 D 2511 4291 D 2520 4330 D 2529 4369 D 2539 4408 D 2548
4448 D 2558 4487 D 2569 4526 D 2579 4565 D 2590 4604 D 2602 4643 D 2613 4681 D
2626 4720 D 2639 4758 D 2653 4795 D 2668 4833 D 2686 4870 D 2705 4905 D 2728
4938 D 2757 4966 D 2795 4979 D 2833 4966 D 2862 4938 D 2886 4905 D 2906 4870 D
2925 4835 D 2942 4798 D 2957 4760 D 2973 4723 D 2987 4685 D 3001 4648 D 3014
4609 D 302

7 4571 D 3040 4533 D 3052 4494 D 3065 4456 D 3077 4417 D 3089 4379 D 3101 4340
D 3113 4302 D 3125 4263 D 3137 4225 D 3149 4186 D 3162 4148 D 3173 4109 D 3185
4071 D 3197 4032 D 3210 3994 D 3222 3955 D 3234 3917 D 3247 3878 D 3260 3840 D
3272 3802 D 3286 3764 D 3298 3725 D 3311 3687 D 3325 3649 D 3339 3611 D 3353
3573 D 3367 3535 D 3381 3498 D 3396 3460 D 3411 3423 D 3426 3386 D 3442 3348 D
3458 3311 D 3475 3274 D 3491 3237 D 3508 3201 D 3526 3164 D 3544 3128 D 3562
3092 D 3581 3057 D 3601 3022 D 3621 2986 D 3641 2952 D 3662 2917 D 3685 2883 D
3707 2849 D 3730 2817 D 3754 2784 D 3778 2751 D 3803 2720 D 3829 2689 D 3856
2659 D 3883 2630 D 3911 2600 D 3941 2572 D 3970 2545 D 4001 2518 D 4032 2493 D
4064 2469 D 4097 2445 D 4130 2422 D 4164 2400 D 4198 2379 D 4233 2359 D 4269
2340 D 4305 2322 D 4342 2305 D 4379 2289 D 4416 2273 D 4454 2259 D 4492 2245 D
4530 2232 D 4569 2219 D 4607 2208 D 4646 2197 D 4685 2187 D 4725 2177 D 4764
2168 D 4803 2160 D 4843 2152 D 4882 2144 D 4922 2137 D 4962 2131 D 5002 2124 D
5042 2

118 D 5082 2112 D 5122 2107 D 5162 2103 D 5202 2098 D 5242 2093 D 5283 2089 D
5323 2086 D 5363 2082 D 5403 2079 D 5444 2075 D 5483 2072 D 5524 2070 D 5564
2067 D 5605 2065 D 5645 2062 D 5685 2060 D 5726 2058 D 5766 2056 D 5806 2053 D
5846 2052 D 5887 2051 D 5927 2049 D 5967 2047 D 6008 2046 D 6048 2044 D 6088
2044 D 6129 2042 D 6169 2041 D 6209 2040 D 6250 2039 D S
2050 5900 M 6250 5900 D S
6250 1700 M 6250 5900 D S
showpage
end
grestore


